Author’s response to reviews

Title: Two-stage case-control association study of dopamine-related genes and migraine

Authors:

Roser Corominas (rcoromin@ir.vhebron.es)
Marta Ribases (marta.ribases@gmail.com)
Montserrat Camiña (montse@xpereira.com)
Ester Cuenca-León (ecuenca@ir.vhebron.net)
Julio Pardo (jpardof@meditex.es)
Susana Boronat (sboronat@vhebron.net)
María-Jesús Sobrido (mariajesus.sobrido@usc.es)
Bru Cormand (bcormand@ub.edu)
Alfons Macaya (amacaya@vhebron.net)

Version: 3 Date: 2 September 2009

Author’s response to reviews: see over
Dear Dr. Kaiser,

Thank you for the second review of our manuscript entitled “Two-stage case-control association study of dopamine-related genes and migraine”. We again thank the referee for her useful criticisms. We have now revised the manuscript and made the following changes according to the referee recommendations:

**Referee # 1**

The referee suggests that we remove any statements alluding to the study being “the most complete analysis” of dopamine genes in migraine.

We have deleted the last sentence of the introduction, where we presented our study as a “the first comprehensive study of the contribution of the dopaminergic pathway to migraine susceptibility in the Spanish population”. Any other sentence claiming a primacy of our work in the field of migraine had been removed in the first revision of the manuscript.

We hope that now the manuscript will be in a suitable form for its publication in BMC Medical Genetics.

Sincerely,

Alfons Macaya  MD, PhD